Author's response to reviews

Title: FREQUENCY PATTERNS OF CORE CONSTIPATION SYMPTOMS AMONG THE ASIAN ADULTS: A SYSTEMATIC REVIEW

Authors:

Patimah Abdul Wahab (patimahabdul@gmail.com)

Yeong Yeh Lee (justnleeyy@gmail.com)

Dariah Mohd Yusoff (dariah@usm.my)

Version: 2 Date: 01 Oct 2017

Author’s response to reviews:

Dear Editor,

Based on the comments, we have changed the authors' full names to the authors' initials in the declaration section (line 311, page 13) and provided a clean copy of the manuscript.

Thank you.

Kind Regards,

Patimah